Author’s response to reviews

Title: Patients affected with Fabry disease have an increase incidence of progressive hearing loss and sudden deafness: an investigation of twenty-two hemizygous male patients

Authors:

Dominique P Germain (dominique.germain@hop.egp.ap-hop-paris.fr)
Paul Avan (paul.avan@u-clermont1.fr)
Augustin Chassaing (augustin.chassaing@hop.egp.ap-hop-paris.fr)
Pierre Bonfils (pierre.bonfils@hop.egp.ap-hop-paris.fr)

Version: 3 Date: 10 Oct 2002

The title of the 'Patients and Methods' section has been changed to "Methods"
Figures have all been cropped
The Y axis legend of figure 4 has been modified
The references have been carefully checked for typographical errors